Author's response to reviews

Title: Arrhythmogenic substrate at interventricular septum as a target site for radiofrequency catheter ablation of recurrent ventricular tachycardia in left dominant arrhythmogenic cardiomyopathy

Authors:

Stepan Havranek (stepan.havranek@lf1.cuni.cz)
Tomas Palecek (tomas.palecek@vfn.cz)
Ivana Vitkova (ivana.vitkova@vfn.cz)
Dan Wichterle (wichterle@hotmail.com)

Version: 2
Date: 27 November 2014

Author's response to reviews: see over
November 27, 2014

Professor Timothy Shipley.
Executive Editor

We have resubmitted our manuscript entitled "Arrhythmogenic substrate at interventricular septum as a target site for radiofrequency catheter ablation of recurrent ventricular tachycardia in left dominant arrhythmogenic cardiomyopathy". Manuscript category: Case report. Manuscript number: MS: 201806555151183

We request for review our manuscript by the editorial committee of the BMC Cardiovascular Disorders.

Thank you for your recent response to our first submission. According to your advice we corrected manuscript appropriately. We added line and page numbering and we completed CARE Checklist.

This report describes a case of left dominant arrhythmogenic cardiomyopathy manifested with tolerated sustained monomorphic ventricular tachycardia treated by catheter ablation.

All authors have made substantial contributions to drafting the article or revising it critically for important intellectual content and gave final approval of the version to be submitted.

We state that the case report is original, is not under simultaneous consideration by other journals and has not yet been published either wholly or in part, except as a conference abstract. The authors report no financial relationships or conflicts of interest regarding the content herein.

Thank you very much for your consideration.

Sincerely,

Stepan Havranek, M.D., corresponding author
2nd Internal Department - Department of Cardiology and Angiology
1st Faculty of Medicine, Charles University
U Nemocnice 2, Prague, 128 08, Czech Republic
Tel. +420 224962605, Fax +420 224912611, e-mail: stepan.havranek@lf1.cuni.cz